Author's response to reviews

Title: Suitability of measures of self-reported medication adherence for routine clinical use: A systematic review.

Authors:

Sara Garfield (saragarfield@yahoo.co.uk)
Sarah Clifford (sarah.clifford@pharmacy.ac.uk)
Lina Eliasson (l.eliasson@imperial.ac.uk)
Nick Barber (n.barber@pharmacy.ac.uk)
Alan Willson (alan.willson@nliah.wales.nhs.uk)

Version: 4 Date: 2 November 2011

Author's response to reviews: see over
Dear Sir/Madam,

Re MS: 1290251559540329 Suitability of adherence measures for routine clinical practice: A systematic review

Further to your email of 01/11/11 we have made the changes requested. A point by point description is given below:

Major revisions:

1) We have altered the author list in the submission system so that it matches the manuscript.
2) We have reformatted the abstract as requested.
3) We have added a title for figure 1 and added it to the manuscript file.
4) We have added an "Additional files" section at the end of the manuscript. Both additional files remaining in the system are mentioned in the text.

Minor revisions:

1) We have reformatted the email addresses as requested.
2) We have removed the term ‘et al’ from the reference list and added in all authors.
3) We have checked the text for typographical errors.

Yours faithfully,

Dr Sara Garfield.  Postdoctoral Research Fellow, The School of Pharmacy, University of London
**Professor Nick Barber.** The School of Pharmacy, University of London, Visiting Professor in Patient Safety, Harvard Medical School

**Dr Sarah Clifford.** The School of Pharmacy, University of London

**Dr Lina Eliasson.** Imperial College London, University of London

**Dr Alan Willson.** Director of Research and Development, National Leadership and Innovation Agency for Healthcare